Author’s response to reviews

Title: Prognostic nomogram for inpatients with asthma exacerbation

Authors:

Wakae Hasegawa (w-hasegawa@umin.ac.jp)
Yasuhiro Yamauchi (YAMAUCHIY-INT@h.u-tokyo.ac.jp)
Hideo Yasunaga (yasunagah-tky@umin.ac.jp)
Hideyuki Takeshima (Htakeshima-tky@umin.ac.jp)
Yukiyo Sakamoto (ysakamoto-tmd@umin.org)
Taisuke Jo (jo-taisuke@umin.ac.jp)
Yusuke Sasabuchi (rhapsody77777@gmail.com)
Hiroki Matsui (ptmatsui-tky@umin.ac.jp)
Kiyohide Fushimi (kfushimi.hci@tmd.ac.jp)
Takahide Nagase (takahide-tky@umin.ac.jp)

Version: 4 Date: 31 May 2017

Author’s response to reviews:

Dear Helen Armes,

BMC Pulmonary Medicine.

May 31, 2017

Dear Helen Armes,

Response to the Editors’ comments: Manuscript ID: PULM-D-16-00290R3 ‘Prognostic nomogram for inpatients with asthma exacerbation’.

We thank you for your careful and comprehensive review of our manuscript. We have asked a professional language editing service to edit our manuscript and revised our manuscript.
We believe that the revised manuscript is now substantially improved. We would be most grateful if you would consider it for possible publication in BMC Pulmonary Medicine.

We look forward to hearing from you at your earliest convenience.

Sincerely yours,

Yasuhiro Yamauchi

Editor Comments:

1. Please have the text edited by a professional language editing service or a native English speaking colleague. There are still a number of issues with grammar, wording, spelling, and/or punctuation that need to be addressed before acceptance for publication.

For example use of the word “ambulatory” where it should read “ambulance” in results section p7 and also table 1

> Thank you for your comments. In accordance with your recommendation, we have asked a professional language editing service to edit our manuscript, and we have revised our manuscript.